Author's response to reviews

Title: A qualitative exploration of the emotional wellbeing and support needs of new mothers from Afghanistan living in Melbourne, Australia

Authors:

Alana Russo (alana.russo@burnet.edu.au)
Belinda Lewis (belinda.lewis@monash.edu)
Andrew Joyce (ajoyce@swin.edu.au)
Belinda Crockett (bcrockett@swin.edu.au)
Stanley Luchters (sluchters@burnet.edu.au)

Version: 4 Date: 2 July 2015

Author's response to reviews: see over
2th July, 2015

TO: Nawsheen Boodhun, Executive Editor, BMC Pregnancy and Childbirth

Re: ‘A qualitative exploration of the emotional wellbeing and support needs of new mothers from Afghanistan living in Melbourne, Australia’.

Dear Nawsheen, Ian, and Co-editors,

Thank you for the opportunity to revise and submit our manuscript to BMC Pregnancy and Childbirth. As requested by the editor, the following amendments have been made:

- Further information has been provided about the respondent validation process. Please refer to line 231 – 238 of the manuscript.
- All bold font has been removed from the reference list.

We appreciate the time and support provided by the editors throughout this process. We look forward to hearing the outcome of the review process.

Yours sincerely,

Alana Russo
Burnet Institute